Author's response to reviews

Title: A web-based simulation of a longitudinal clinic used in a 4-week ambulatory rotation: a cohort study

Authors:

Rene Wong (Rene.Wong@uhn.on.ca)
Heather Lochnan (hlochnan@ottawahospital.on.ca)

Version: 4 Date: 9 January 2009

Author's response to reviews:

Rene Wong, MD, FRCPC
200 Elizabeth Street, 12-EN-211
Toronto General Hospital
Toronto, Ontario, Canada M5G 2C4
416-340-4053 (P), 416-340-3314 (F)
Rene.Wong@uhn.on.ca

January 9, 2009

BMC Medical Education
Attention: Editors

To Whom It May Concern:

Re: MS: 1640056089221574
A web-based simulation of a longitudinal clinic used in a 4-week ambulatory rotation: a cohort study
Authors: Rene Wong and Heather Lochnan

Thank you for the review of this manuscript and for the additional comments provided by the two reviewers. This most recent revision of the manuscript has incorporated the most recent suggestions by the reviewers. Specifically, in the “Results” section, numeric results have been incorporated into the text to complement table 4. Minor grammatical changes have also been made to the ‘Results’ section and Table 5 as suggested by Ms. Michaelsen.

If there are any questions about this report, please do not hesitate to contact me.
Sincerely yours,

Rene Wong, MD, FRCPC